Author's response to reviews

Title: Building on the EGIPPS performance assessment: A multipolar framework as a heuristic to tackle the complexity of performance of public service oriented health care organisations

Authors:

Bruno Marchal (bmarchal@itg.be)
Tom Hoërée (thoeree@itg.be)
Valéria Campos da Silveira (vcdsilveira@itg.be)
Sara Van Belle (sara.vanbelle@lshtm.ac.uk)
Nuggehalli S. Prashanth (prashanthns@iphindia.org)
Guy Kegels (gkegels@itg.be)

Version: 4 Date: 31 March 2014

Author's response to reviews: see over
Authors’ response to the comments of the Editors

We would like to thank the peer reviewer and the editors for their constructive remarks.

In response to the editorial comments, we inserted the following sections at the end of the paper:

Competing interests

The authors declare that they have no competing interests.

Author contributions

BM, TH and GK contributed to the conception and design of the paper. BM, TH, VcDS, SVB, NSP and GK were involved in drafting of the manuscript and its revisions. BM edited the final draft. All authors read and approved the final manuscript. All authors agree to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.

Acknowledgements

We would like to thank all the participants of the Option/Short Course in Strategic Management of Health Systems, organised yearly at the Institute of Tropical Medicine, Antwerp since 2007. Their struggle with the organizational performance assessment assignment helped us improve the multipolar performance framework.